ORIGINAL ARTICLE

Empagliflozin Improves Kidney Outcomes in Patients With or Without Heart Failure Insights From the EMPA-REG OUTCOME Trial

BACKGROUND: In EMPA-REG OUTCOME (Empagliflozin Cardiovascular Outcome Event Trial in Type 2 Diabetes Mellitus Patients) empagliflozin significantly reduced the risk of cardiovascular and kidney outcomes in patients with type 2 diabetes mellitus and established cardiovascular disease. Post hoc, we evaluated empagliflozin on kidney outcomes in patients with or without heart failure (HF).

METHODS AND RESULTS: Individuals were randomized to empagliflozin 10 mg, 25 mg, or placebo. Prespecified analyses by baseline HF status included risk of incident or worsening nephropathy and estimated glomerular filtration rate slope analyses. Cox proportional hazards models assessed consistency of treatment effect across subgroups. Safety evaluations included kidney-related adverse events. At baseline, 244 (10.5%) and 462 (9.9%) patients had HF in the placebo and empagliflozin groups, respectively. Overall, the incidence of kidney outcome events was numerically higher in patients with than without HF. In the HF group, empagliflozin reduced risk of incident or worsening nephropathy or cardiovascular death by 43% (hazard ratio, 0.57 [95% CI, 0.42–0.77]) and progression to macroalbuminuria by 50% (hazard ratio, 0.50 [0.33–0.75]). After an initial transient decrease, estimated glomerular filtration rate stabilized over time with empagliflozin but gradually declined with placebo. Kidney effects in patients with HF were consistent with those in the overall study population (all P values for interaction >0.05). Across groups, the incidence rate of kidney-related adverse events/100 patient-years was higher in patients with than without HF; however, overall rates were comparable between groups.

CONCLUSIONS: These findings from EMPA-REG OUTCOME support the hypothesis that empagliflozin could reduce the risk of clinically relevant kidney events and may slow progression of chronic kidney disease in individuals with type 2 diabetes mellitus regardless of HF status.

CLINICAL TRIAL REGISTRATION: URL: https://www.clinicaltrials.gov. Unique identifier: NCT01131676. Javed Butler, MD, MPH, MBA Faiez Zannad, MD, PhD David Fitchett, MD Bernard Zinman, MD Audrey Koitka-Weber, PhD Maximilian von Eynatten, MD Isabella Zwiener, PhD Jyothis George, MBBS, PhD, FRCP Martina Brueckmann, MD Alfred K. Cheung, MD Christoph Wanner, MD

Key Words: cardiovascular diseases chronic kidney disease
diabetes mellitus, type 2
heart failure sodium-glucose transporter 2

© 2019 American Heart Association, Inc.

https://www.ahajournals.org/journal/ circheartfailure

WHAT IS NEW?

- Chronic kidney disease is a common condition and predicts a worse prognosis in patients with heart failure (HF), especially in those with type 2 diabetes mellitus.
- Patients with type 2 diabetes mellitus and HF are at higher risk of adverse kidney outcomes compared with patients without HF.
- In a subgroup analysis of the EMPA-REG OUTCOME study (Empagliflozin Cardiovascular Outcome Event Trial in Type 2 Diabetes Mellitus Patients), the SGLT2 (sodium-glucose cotransporter 2) inhibitor empagliflozin reduced the risk of clinically relevant kidney events and slowed progression of chronic kidney failure in patients with and without HF at baseline. These effects were consistent with those reported for the overall study population.

WHAT ARE THE CLINICAL IMPLICATIONS?

- Empagliflozin may have the potential to slow progressive kidney function loss in patients with type 2 diabetes mellitus and concomitant HF.
- Several randomized clinical trials are now underway that will investigate further the potential role of SGLT2 inhibitors as a treatment specifically for HF.
- These trials are enrolling HF patients with or without type 2 diabetes mellitus, and those with an estimated glomerular filtration rate as low as 20 mL/min per 1.73 m².

eart failure (HF) is a major global problem, affecting ≈26 million people worldwide in 2014.1 In developed countries, HF prevalence is ≈1% to 2% of the adult population, increasing to $\geq 10\%$ in people aged >70 years.² Hospitalization for HF accounts for over 1 million admissions per year in both Europe and the United States.¹ Type 2 diabetes mellitus (T2DM) is common in people with HF with numerous studies showing a prevalence of up to 50%.³ T2DM is associated with a 2- to 5-fold increased risk of developing HF^{2,3} as well as a worse prognosis in those with concomitant presence of both as opposed to either one alone.⁴ The presence of T2DM is also associated with a higher rate of, and worse outcomes with, hospitalization for HF.⁵ A commonly linked comorbidity with HF and T2DM is chronic kidney disease (CKD), which affects over 10% of the population in many countries. Approximately 40% of people with HF have concomitant impairment of kidney function (defined as an estimated glomerular filtration rate [eGFR] <60 mL/min per 1.73 m²).⁶ Similar to T2DM and HF associations, the presence of CKD is common in either or both conditions and has an overlapping, underlying common pathophysiology, and frequently worsens prognosis.

Empagliflozin is a selective SGLT2 (sodium-glucose cotransporter 2) inhibitor indicated for the treatment of T2DM. In the EMPA-REG OUTCOME trial (Empagliflozin Cardiovascular Outcome Event Trial in Type 2 Diabetes Mellitus Patients), empagliflozin, when given in addition to standard of care and compared with placebo, significantly reduced the risk of cardiovascular death by 38%,⁷ improved clinically relevant kidney outcomes,⁸ and slowed the progression of kidney function decline,⁸ in patients with T2DM and established cardiovascular disease. The beneficial cardiovascular effects of empagliflozin included a reduction in the risk for hospitalization for HF and cardiovascular death, with consistent effects seen in patients with or without HF at baseline.⁹

CKD and HF have common interrelated underlying pathophysiology (eg, inflammation, comorbidity burden, hemodynamic changes, and neurohormonal activation), and the progression of CKD and its associated adverse outcomes in people with T2DM may differ in people with or without HF. Importantly, the glucoselowering drug empagliflozin induces secondary diuresis and natriuresis, with consequent hemodynamic influences on the kidney, at the arteriolar level. These mechanisms could be important for the long-term progression of both CKD and HF; however, their implications are still poorly understood. Here, we report analyses from the EMPA-REG OUTCOME trial investigating post hoc the impact of empagliflozin on prespecified kidney outcomes in patients with or without HF at baseline.

METHODS

Data Sharing

To ensure independent interpretation of clinical study results, Boehringer Ingelheim grants all external authors access to all relevant material, including participant-level clinical study data, and relevant material as needed by them to fulfill their role and obligations as authors under the ICMJE criteria. Furthermore, clinical study documents (eg, study report, study protocol, and statistical analysis plan) and participant clinical study data are available to be shared after publication of the primary article in a peer-reviewed journal and if regulatory activities are complete and other criteria met per the BI Policy on Transparency and Publication of Clinical Study Data: https://trials.boehringer-ingelheim.com/ transparency_policy.html.

Before providing access, documents will be examined, and, if necessary, redacted and the data will be de-identified, to protect the personal data of study participants and personnel, and to respect the boundaries of the informed consent of the study participants. Clinical study reports and related clinical documents can be requested via this link: https://trials.boehringer-ingelheim.com/trial_results/clinical_submission_documents.html.

All such requests will be governed by a Document Sharing Agreement. Bona fide, qualified scientific, and medical researchers may request access to de-identified, analyzable participant clinical study data with corresponding documentation describing the structure and content of the data sets. On approval, and governed by a Data Sharing Agreement, data are shared in a secured data-access system for a limited period of 1 year, which may be extended upon request. Researchers should use https://clinicalstudydatarequest.com to request access to study data.

Trial Design and Procedures

The design of the EMPA-REG OUTCOME trial has been described previously.^{4,7,9} Briefly, patients with T2DM (with glycated hemoglobin [HbA1c] 7.0%–9.0% for drug-naïve patients and 7.0%–10.0% for those on stable glucose-lowering therapy) and established cardiovascular disease were eligible to participate. Individuals were required to have a body mass index (BMI) of \leq 45 kg/m² and eGFR >30 mL/min per 1.73 m² at screening (as calculated using the modification of diet in renal disease formula). Patients were randomized 1:1:1 to once daily empagliflozin 10 mg, 25 mg, or placebo, in addition to standard of care.

The trial was planned to continue until at least 691 patients experienced a composite primary outcome event (defined as first occurrence of cardiovascular death, nonfatal myocardial infarction, or nonfatal stroke). Cardiovascular outcome events and deaths were prospectively adjudicated by Clinical Events Committees.⁷ Patients were asked to attend the clinic at prespecified times during the study including a follow-up visit 30 days following the end of the treatment period. Patients who prematurely discontinued their study medication were asked to attend all visits as originally planned.¹⁰

For laboratory values, serum creatinine and urinary albuminto-creatinine ratio in spot urine samples obtained during regular study visits were determined in central laboratories using standardized procedures. The trial was conducted in accordance with the principles of the Declaration of Helsinki and the International Conference on Harmonization Good Clinical Practice guidelines and was approved by local authorities. An independent ethics committee or institutional review board approved the clinical protocol at each participating center. Patients provided written informed consent before study entry.

Heart Failure

Patients with HF were allowed to participate without any restriction about ejection fraction or New York Heart Association functional class symptoms. HF was neither required nor excluded for participation. Investigator-reported HF status at study baseline was identified by searching for preferred terms in the Standardized Medical Dictionary for Drug Regulatory Activities Query "cardiac failure" (comprising acute pulmonary edema; cardiac failure; cardiac failure, acute; cardiac failure, chronic; cardiac failure, congestive; cardiogenic shock; cardiopulmonary failure; left ventricular failure; pulmonary edema; and right ventricular failure).⁹ Cardiac imaging was not performed or required for the definition of HF.

Kidney Outcomes

Results of the primary outcomes and secondary outcomes from the EMPA-REG OUTCOME trial have been previously reported in the overall population.⁸ Prespecified kidney outcomes included incident or worsening nephropathy, defined as a composite of progression to macroalbuminuria (urine albumin-to-creatinine ratio, >300 mg/g); a doubling of serum creatinine level accompanied by an eGFR value of ≤45 mL/min per 1.73 m² (modification of diet in renal disease); initiation of renal replacement therapy; or death from renal disease.⁸ Other prespecified outcomes included a composite of incident or worsening nephropathy or death from cardiovascular causes and the individual components of incident or worsening nephropathy. A previous post hoc analysis also examined the composite of doubling of serum creatinine, initiation of renal replacement therapy, or death from renal disease.⁸ The post hoc analyses presented herein focused on the outcomes mentioned above in the specific patient subgroups with or without investigator-reported HF at baseline. We also assessed a composite of sustained (ie, ≥2 consecutive measurements that were \geq 4 weeks apart) eGFR decline of \geq 40%, initiation of renal replacement therapy, or renal death.

Changes in eGFR values were assessed over time, alongside a prespecified eGFR slope analysis for 3 prespecified study periods (treatment-initiation effects from baseline to week 4; chronic maintenance treatment effects from week 4 to last value on treatment; and post-treatment effects from last value on treatment to follow-up).¹¹

Furthermore, progression to sustained improvement or deterioration of albuminuria category, in terms of progression to sustained normoalbuminuria or microalbuminuria in patients with macroalbuminuria at baseline (as a measure of improvement in kidney function) and progression to sustained macroalbuminuria in patients with normoalbuminuria or microalbuminuria at baseline (as a measure of deterioration) were also assessed.¹²

Safety was assessed based on adverse events (AEs) reported in subgroups by HF status. Throughout this analysis, AE data were based on investigator reports without any formal adjudication.

Statistical Analysis

For efficacy outcomes, we used a modified intent-to-treat approach to perform analyses in patients treated with ≥ 1 dose of study drug (treated set) and compared the placebo and pooled empagliflozin (10 and 25 mg) groups. The modification of diet in renal disease formula was used to calculate eGFR at baseline and for all eGFR-based analyses (including slopes). For kidney outcomes, Kaplan-Meier estimates and a Cox proportional hazards model was used to investigate the consistency of the treatment effect between the subgroup of patients with versus without HF at baseline. The Cox model included treatment, age, sex, baseline BMI category, baseline HbA1c category, baseline eGFR, geographic region, prior HF, and treatment-by-prior-heart-failure interaction.

A mixed model, repeated measures analysis was used to evaluate changes from baseline in eGFR over time and included patients who had a baseline and postbaseline eGFR measurement, using all available data, including values obtained on treatment, post-treatment, and after intake of rescue medication. The model included baseline eGFR and baseline HbA1c as linear covariates and geographic region, baseline BMI category, the last week a patient could have had an eGFR measurement, treatment, visit, HF at baseline, treatment by visit interaction, visit by HF at baseline interaction, treatment by HF at baseline interaction, treatment by visit by HF at baseline interaction, baseline HbA1c by visit interaction, and baseline eGFR by visit interaction as fixed effects. Because measurements of the variable of interest collected from the same patient at different time points are correlated, these data constitute the repeated measures in the mixed model, repeated measures model. Calculation of eGFR slopes within the 3 prespecified study periods was performed by applying a separate random coefficient model for each period allowing for random intercept random slope per patient.¹¹ The model included baseline HbA1c as a linear covariate, and geographic region, baseline BMI category, heart-failure-at-baseline-by-treatment interaction, and time-by-heart-failure-at-baseline-by-treatment as fixed classification effects. Patients were required to provide at least 2 measurements per study period to be included in the respective analysis, that is, baseline and week 4 measurements for the acute period, and last value on treatment and follow-up measurements for the post-treatment period. Analysis of the treatment-initiation period assumed that any patients lost to follow-up or otherwise excluded were missing completely at random. Analyses for the chronic maintenance and post-treatment periods assumed that any patients lost to follow-up or otherwise excluded before this specific study period were also missing completely at random. In addition, analyses of the chronic maintenance treatment period required that missing data occurring after the start of the chronic phase followed a missing at random mechanism—this includes missing data because of (1) lost to follow-up and (2) artificial censoring when patients discontinued study medication. The missing at random assumption allows the probability of missing data to depend on the observed covariates and the observed values for eGFR but not on the values of eGFR that were not observed.11

AEs were assessed descriptively and presented separately for the empagliflozin 10 and 25 mg groups. Exposureadjusted incidence rates were calculated and included events until 7 days after the last receipt of the study drug. All analyses were performed using SAS version 9.4.

RESULTS

In total, 97.0% of patients completed the study, with 25.4% of patients prematurely discontinuing study medication, and final vital status available for 99.2% of patients.⁷

Baseline Characteristics

At baseline, 244 (10.5%) and 462 (9.9%) individuals in the placebo and empagliflozin group, respectively, had investigator-reported HF. Baseline characteristics for patients with or without HF have been previously published in detail⁹; selected key characteristics are shown in Table 1. Baseline characteristics were generally balanced between the empagliflozin and placebo groups, although some differences were noted between patients with versus without HF. Thus, patients with HF were slightly older, had a higher BMI, and a greater percentage of patients had a history of myocardial infarction or atrial fibrillation, and an eGFR <60 mL/min per 1.73 m². As expected, more patients with HF were receiving β -blockers, diuretics, and mineralocorticoid receptor antagonists before randomization (Table 1).

Kidney Outcomes

Patients with HF were at overall greater risk of CKD progression versus those without HF. Thus, the incidence rate of kidney outcome events was numerically higher in patients with concomitant HF (Figure 1). Empagliflozin significantly reduced the risk of clinically relevant kidney outcome events as compared with placebo in both patients with and without HF (all *P* values for interaction >0.05; Figure 1; Figure I in the Data Supplement), and this observation was consistent with findings in the overall population.⁸ Similar findings were also seen when all-cause mortality was added in the composite outcome (Figure III in the Data Supplement).

Kidney Function

Figure 2 depicts changes in eGFR over the course of the study. The overall number and percentage of patients included in this analysis were N=6967 and 99.2% of the total population, respectively. Patients with HF had an overall lower baseline eGFR compared with those without HF. Despite this difference in kidney function at baseline, a consistent pattern in eGFR change over time was apparent independent of HF status, which was characterized by an acute decline in eGFR at week 4, followed by a period of stable kidney function during long-term follow-up. This observation was consistent with findings in the overall population.⁸

Adjusted mean eGFR slopes during the 3 prespecified study periods are shown in Figure II in the Data Supplement. During the first 4 weeks of treatment, the weekly mean adjusted eGFR decrease was numerically larger in the empagliflozin compared with the placebo groups (for both subgroups with or without HF; Figure IIA in the Data Supplement). Thereafter, the annual adjusted change in mean eGFR during the chronic maintenance treatment period was stable in both empagliflozin subgroups but declined in the placebo subgroups (Figure IIA in the Data Supplement). During the post-treatment follow-up, the weekly adjusted mean eGFR increased and returned towards mean baseline eGFR levels in the empagliflozin subgroups, whereas there was little change seen in eGFR levels in the placebo groups (Figure IIA in the Data Supplement). As indicated by P values for interaction (all >0.05), the pattern of eGFR changes was consistent in the subgroups of patients

Table 1. Baseline Characteristics of Patients With and Without HF at Baseline

	Patients W	ith HF* at Baseline	Patients Without HF* at Baseline						
	Placebo (N=244)	Empagliflozin (N=462)	Placebo (N=2089)	Empagliflozin (N=4225)					
Age, mean (SD), y	64.5 (8.9)	64.5 (8.8)	63.1 (8.8)	63.0 (8.5)					
Male, n (%)	175 (71.7)	320 (69.3)	1505 (72.0)	3016 (71.4)					
eGFR, mean (SD), mL/min per 1.73 m ²	69.3 (20.7)	68.4 (20.2)	74.3 (21.0)	74.8 (21.6)					
eGFR, n (%)									
≥90, mL/min per 1.73 m ²	37 (15.2)	73 (15.8)	451 (21.6)	977 (23.1)					
60–<90, mL/min per 1.73 m ²	118 (48.4)	215 (46.5)	1120 (53.6)	2208 (52.3)					
<60, mL/min per 1.73 m ²	89 (36.5)	174 (37.7)	518 (24.8)	1038 (24.6)					
Urine albumin-to-creatinine ratio, n (%)									
<30 mg/g	140 (57.4)	272 (58.9)	1242 (59.5)	2517 (59.6)					
30–300 mg/g	72 (29.5)	130 (28.1)	603 (28.9)	1208 (28.6)					
>300 mg/g	31 (12.7)	31 (12.7) 59 (12.8)		450 (10.7)					
Body mass index, mean (SD), kg/m ²	32.3 (5.4)	31.9 (5.6)	30.5 (5.2)	30.5 (5.2)					
Cardiovascular risk factor, n (%)	243 (99.6)	461 (99.8)	2064 (98.8)	4196 (99.3)					
Coronary artery disease	209 (85.7)	385 (83.3)	1554 (74.4)	3160 (74.8)					
Multivessel coronary artery disease	122 (50.0)	226 (48.9)	978 (46.8)	1953 (46.2)					
History of myocardial infarction	164 (67.2)	292 (63.2)	919 (44.0)	1898 (44.9)					
Coronary artery bypass graft	70 (28.7)	119 (25.8)	493 (23.6)	1056 (25.0)					
History of stroke†	57 (23.4)	108 (23.4)	496 (23.7)	976 (23.1)					
Peripheral artery disease	50 (20.5)	100 (21.6)	429 (20.5)	882 (20.9)					
Atrial fibrillation	43 (17.6)	76 (16.5)	99 (4.7)	171 (4.0)					
Single vessel coronary artery diseaset	26 (10.7)	45 (9.7)	212 (10.1)	453 (10.7)					
HbA1c, mean (SD), %	8.01 (0.82)	8.11 (0.87)	8.09 (0.85)	8.06 (0.85)					
Antihypertensive therapy, n (%)	242 (99.2)	456 (98.7)	1979 (94.7)	3990 (94.4)					
ACE inhibitors/angiotensin receptor blockers	206 (84.4)	406 (87.9)	1662 (79.6)	3392 (80.3)					
β-blockers	199 (81.6)	360 (77.9)	1299 (62.2)	2696 (63.8)					
Diuretics	172 (70.5)	334 (72.3)	816 (39.1)	1713 (40.5)					
Loop diuretics	110 (45.1)	224 (48.5)	254 (12.2)	501 (11.9)					
Calcium channel blockers	70 (28.7)	111 (24.0)	718 (34.4)	1418 (33.6)					
Mineralocorticoid receptor antagonists	53 (21.7)	116 (25.1)	83 (4.0)	189 (4.5)					
Renin inhibitors	1 (0.4)	3 (0.6)	18 (0.9)	24 (0.6)					
Other	20 (8.2)	33 (7.1)	171 (8.2)	350 (8.3)					
Systolic BP, mean (SD), mmHg	134.9 (19.2)	133.6 (16.9)	135.9 (17.0)	135.4 (16.9)					
Diastolic BP, mean (SD), mm Hg	76.4 (10.6)	76.6 (10.2)	76.9 (10.1)	76.6 (9.7)					

Data in patients treated with at least one dose of study drug. eGFR: empagliflozin n=4223 for patients without HF at baseline. UACR: placebo n=243 and empagliflozin n=461 for patients with HF at baseline; placebo n=2074 and empagliflozin n=4175 for patients without HF at baseline. ACE indicates angiotensinconverting enzyme; BP, blood pressure; eGFR, estimated glomerular filtration rate; HbA1c, glycated hemoglobin; HF, heart failure; MedDRA, Medical Dictionary for Regulatory Activities; and UACR, urine albumin-to-creatinine ratio.

*Based on narrow Standardized MedDRA Query (SMQ) cardiac failure, which comprised these preferred terms: acute pulmonary edema; cardiac failure; cardiac failure, acute; cardiac failure, chronic; cardiac failure, congestive; cardiogenic shock; cardiopulmonary failure; left ventricular failure; pulmonary edema; and right ventricular failure.

†Information was not available for 1 patient.

with or without HF, and the findings were consistent with those reported in the overall population.¹¹

Albuminuria

Figure 3 shows changes in albuminuria over the course of the study. Fewer patients in the empagliflozin versus

placebo groups had a deterioration in albuminuria category, defined as progression to sustained macroalbuminuria in patients with normoalbuminuria or microalbuminuria at baseline. This outcome was consistent across the subgroups of patients with or without HF (*P* value for interaction, *P*=0.7957), and the findings were consistent with those reported in the overall population.¹²

	Empaglif	Empagliflozin		00			p-value for
	n/N	%	n/N	%	HR† (95% CI)	HR† (95% CI)	interaction
Incident or worsening nephr	opathy* or CV de	eath					1
All patients	675/4170	16.2	497/2102	23.6	0.61 (0.55, 0.69)	⊷	
HF at baseline: NO	579/3758	15.4	425/1887	22.5	0.62 (0.55, 0.70)		0.5869
HF at baseline: YES	96/412	23.3	72/215	33.5	0.57 (0.42, 0.77)		
Incident or worsening nephr	opathy*						
All patients	525/4124	12.7	388/2061	18.8	0.61 (0.53, 0.70)	⊷	
HF at baseline: NO	464/3723	12.5	339/1855	18.3	0.62 (0.54, 0.71)		0.4152
HF at baseline: YES	61/401	15.2	49/206	23.8	0.53 (0.36, 0.77)		
Progression to macroalbum	inuria** (UACR >	300 mg	g/g)			I	
All patients	459/4091	11.2	330/2033	16.2	0.62 (0.54, 0.72)	⊷ •••	
HF at baseline: NO	411/3697	11.1	289/1828	15.8	0.64 (0.55, 0.75)		0.2551
HF at baseline: YES	48/394	12.2	41/205	20.0	0.50 (0.33, 0.75)		
Doubling of serum creatinin initiation of renal replacement	e level with eGFF nt therapy, or de	R ≤45 m ath froi	nl/min/1.73 r m renal dise	n², ease		:	
All patients	81/4645	1.7	71/2323	3.1	0.54 (0.40, 0.75)		
HF at baseline: NO	64/4187	1.5	60/2082	2.9	0.50 (0.35, 0.72)	· · · · · · · · · · · · · · · · · · ·	0.3097
HF at baseline: YES	17/458	3.7	11/241	4.6	0.78 (0.36, 1.67)		
Sustained [‡] eGFR decline of a or death from renal disease	≥40%, initiation o	of renal	replaceme	nt thera	ipy,		
All patients	100/4645	2.2	86/2323	3.7	0.55 (0.41, 0.73)	····•	
HF at baseline: NO	79/4187	1.9	72/2082	3.5	0.51 (0.37, 0.70)		0.2739
HF at baseline: YES	21/458	4.6	14/241	5.8	0.78 (0.39, 1.53)	, i	
					0.25	0.5	1 2
					0.23	. 0.0	·

Figure 1. Effects of empagliflozin on kidney outcomes in patients with or without heart failure (HF) at baseline.

CV indicates cardiovascular; and HR, hazards ratio. *Progression to macroalbuminuria (urinary albumin-to-creatinine ratio [UACR] >300 mg/g); a doubling of the serum creatinine level, accompanied by an estimated glomerular filtration rate (eGFR) of \leq 45 mL/min per 1.73 m²; the initiation of renal replacement therapy; or death from renal disease. **Time to macroalbuminuria was assessed only in patients without macroalbuminuria at baseline. †Based on Cox regression analysis in patients who received at least one dose of study drug. ‡Sustained requires 2 consecutive measurements fulfilling the condition that are at least 4 wk apart. eGFR calculated according to Modification of Diet in Renal Disease formula.

Safety

AEs for the overall trial population,⁷ as well as in patient subgroups based on HF status,⁹ have been previously reported. Overall, percentages of patients with AEs, serious AEs, and AEs leading to discontinuation were similar in the empagliflozin and placebo groups. There again, genital infections were more common in patients receiving empagliflozin versus placebo.⁷

Compared with patients without HF at baseline, a higher proportion of patients with HF at baseline had severe AEs, AEs leading to discontinuation, and serious AEs (including fatal AEs) across both treatment groups.



Figure 2. Estimated glomerular filtration rate (eGFR) over time.

N at baseline=6967 (99.2% of total population): patients with at least a baseline and one postbaseline measurement of eGFR. Mixed model, repeated measures analysis in patients who received at least one dose of study drug. HF indicates heart failure.





HF indicates heart failure; and HR, hazard ratio. *Number of patients with normoalbuminuria or microalbuminuria at baseline who were assessable for this end point.

In the current analysis, we further assessed clinically relevant kidney-related AEs. Overall, the rate of kidney-related AEs per 100 patient-years was higher in patients with than without HF, including events reported as acute kidney failure, acute kidney injury, edema, hyperkalemia, volume depletion, and hypotension (Table 2) across both treatment groups. Notably, however, incidence rates were generally balanced between the empagliflozin and placebo groups, respectively. Moreover, events consistent with edema in the overall population were numerically lower in the empagliflozin groups versus the placebo group, (Table 2) and this observation was independent of HF status.

DISCUSSION

In this post hoc analysis from the EMPA-REG OUT-COME trial, patients with T2DM and concomitant HF were at a greater risk of progressive CKD, eventually leading to adverse kidney outcomes, as compared with patients without HF. Although this clinical observation could have been expected based on previous reports of an increased CKD risk in patients with HF, especially if they also have concomitant diabetes mellitus, our report adds significant new insights into the clinical potential of empagliflozin to improve kidney outcomes in this particular vulnerable patient population. Thus, novel analyses based on HF status of study participants revealed that empagliflozin was associated with a significant reduction in risk across various kidney outcomes, including hard kidney outcomes, such as end-stage kidney disease or renal death as well as the additional risk of premature cardiovascular mortality. Notably, the HF status of patients per se did not alter the beneficial kidney effects observed with empagliflozin, and findings in the HF subgroups were overall consistent with those reported in the overall trial population.

Low GFR is often a concern when initiating novel treatments in patients with HF, in particular, if drugs may be prone to worsening short- and long-term kid-

ney function and, therefore, regular assessment of kidney function is recommended in patients with HF. As expected, patients with HF in EMPA-REG OUTCOME had a lower baseline eGFR compared with patients without HF. However, comprehensive eGFR slopes analyses revealed that the pattern of eGFR changes during treatment of up to 178 weeks was similar for both HF subgroups, namely a transient decrease in eGFR with empagliflozin shortly after treatment initiation followed by stabilization during long-term follow-up, while eGFR levels gradually declined in the placebo groups. The pattern of eGFR changes during treatment initiation and long-term follow-up is in line with results published earlier for the overall EMPA-REG population.¹¹ These findings serve as a strong indicator that the proposed hemodynamic effect of empagliflozin, that is, a modulation of the glomerular afferent arteriole tone and concomitant reduction in the glomerular hypertension secondary to diabetes mellitus,^{13,14} is present whether patients have concomitant HF or not. Moreover, the rapid reversal of eGFR after drug cessation in both HF subgroups further indicates that the renal hemodynamic effect of empagliflozin appeared to be swiftly reversible even after long-term treatment. We think these observations are clinically relevant, as they could support physicians in appropriately interpreting shortterm changes in kidney function after starting or stopping empagliflozin, respectively: (1) we did not observe an excess of the initial eGFR dip in patients with HF as compared to those without HF (despite lower eGFR values at treatment initiation of the former) and (2) the rapid recovery of eGFR after drug cessation was not impaired in patients with HF. Most importantly, however, our results suggest that empagliflozin may have the potential to slow progressive kidney function loss in patients with T2DM and concomitant HF.

Albuminuria is another established kidney marker in clinical routine and commonly assessed to stage and monitor kidney damage. Although some drugs indicated for the treatment of patients with HF have also

Table 2. Kidney-Related Adverse Events of Interest in Patients With or Without HF at Baseline

	Placebo (N=2333)		Empagliflozin 10 mg (N=2345)		Empagliflozin 25 mg (N=2342)				
	n/N (%)	Rate*	n/N (%)	Rate*	n/N (%)	Rate*			
Acute kidney failure†									
All patients	155/2333 (6.6)	2.77	121/2345 (5.2)	2.07	125/2342 (5.3)	2.12			
With HF at baseline	21/244 (8.6)	4.12	22/240 (9.2)	4.26	18/222 (8.1)	3.52			
Without HF at baseline	134/2089 (6.4)	2.63	99/2105 (4.7)	1.86	107/2120 (5.0)	1.99			
Acute kidney injury									
All patients	37/2333 (1.6)	0.64	26/2345 (1.1)	0.43	19/2342 (0.8)	0.31			
With HF at baseline	6/244 (2.5)	1.14	7/240 (2.9)	1.28	3/222 (1.4)	0.57			
Without HF at baseline	31/2089 (1.5)	0.59	19/2105 (0.9)	0.35	16/2120 (0.8)	0.29			
Edema									
All patients	216/2333 (9.3)	3.95	106/2345 (4.5)	1.81	106/2342 (4.5)	1.80			
With HF at baseline	30/244 (12.3)	6.17	9/240 (3.8)	1.66	9/222 (4.1)	1.73			
Without HF at baseline	186/2089 (8.9)	3.73	97/2105 (4.6)	1.82	97/2120 (4.6)	1.81			
Hyperkalemia									
All patients	65/2333 (2.8)	1.14	30/2345 (1.3)	0.50	39/2342 (1.7)	0.65			
With HF at baseline	8/244 (3.3)	1.53	5/240 (2.1)	0.91	7/222 (3.2)	1.34			
Without HF at baseline	57/2089 (2.7)	1.10	25/2105 (1.2)	0.46	32/2120 (1.5)	0.58			
Volume depletion									
All patients	115/2333 (4.9)	2.04	115/2345 (4.9)	1.97	124/2342 (5.3)	2.11			
With HF at baseline	17/244 (7.0)	3.31	22/240 (9.2)	4.19	17/222 (7.7)	3.33			
Without HF at baseline	98/2089 (4.7)	1.91	93/2105 (4.4)	1.75	107/2120 (5.0)	2.00			
Hypotension									
All patients	58/2333 (2.5)	1.02	57/2345 (2.4)	0.96	62/2342 (2.6)	1.04			
With HF at baseline	9/244 (3.7)	1.72	9/240 (3.8)	1.65	12/222 (5.4)	2.33			
Without HF at baseline	49/2089 (2.3)	0.95	48/2105 (2.3)	0.89	50/2120 (2.4)	0.92			
Dehydration									
All patients	16/2333 (0.7)	0.28	18/2345 (0.8)	0.30	18/2342 (0.8)	0.30			
With HF at baseline	1/244 (0.4)	0.19	3/240 (1.3)	0.54	1/222 (0.5)	0.19			
Without HF at baseline	15/2089 (0.7)	0.29	15/2105 (0.7)	0.28	17/2120 (0.8)	0.31			

HF indicates heart failure; and MedDRA, Medical Dictionary for Drug Regulatory Activities.

*Rate per 100 patient-years (all events occurred within 7 d after the last receipt of the study drug).

tDecreased renal function, a regulatory term defined by the narrow SMQ 20000003 (acute renal failure), which includes 18 preferred terms (acute prerenal failure, anuria, azotemia, hemodialysis, nephropathy toxic, oliguria, peritoneal dialysis, renal failure, renal failure neonatal, renal impairment neonatal, neonatal anuria, hemofiltration, dialysis, renal impairment, continuous hemodiafiltration, acute kidney injury, acute phosphate nephropathy, and prerenal failure). Hyperkalemia was identified via a search of adverse events by using 2 MedDRA preferred terms (hyperkalemia and increased blood potassium). Edema, an adverse event of special interest, was assessed through a search of adverse events defined using 6 MedDRA preferred terms (fluid overload, fluid retention, generalized edema, edema, edema peripheral, and peripheral swelling).

been shown to reduce levels of albuminuria, such as renin-angiotensin system inhibitors or mineralocorticoid receptor antagonists, others have been associated with an increase in albuminuria after treatment initiation, such as angiotensin receptor-neprilysin inhibitors.¹⁵ Because albuminuria is an established marker of increased kidney risk, including in patients with concomitant HF, treatments that lower rather than increase albuminuria levels may have a plausible pathophysiological basis to improve long-term kidney prognosis. Our analyses showed that fewer patients in the empagliflozin-treated groups showed deterioration in albuminuria, in terms of progression to sustained macroalbuminuria in patients who had normoalbuminuria or microalbuminuria at baseline. This outcome was consistent across the subgroups of patients with or without HF, and similar findings have been reported in the overall patient population.¹² The rapid effect of empagliflozin on reducing albuminuria may at least, in part, be explained by the potential of this drug to lower intraglomerular pressure, as previously reported in individuals with T1DM. The consistent albuminuria-lowering effect of empagliflozin in patients with and without HF in EMPA-REG OUTCOME may further sup-

port a hypothesis that increased rather than decreased intraglomerular pressure (distinct from filtration pressure across the glomerular filtration barrier) may be an important pathophysiological component for progressive kidney function loss in patients with T2DM and concomitant HF. Future mechanistic studies to further decipher the renal hemodynamic and intraglomerular adaptations with SGLT2 inhibitors in patients with HF are, therefore, warranted.

Additional mechanisms beyond renal hemodynamics and reductions in intraglomerular pressure have been proposed to explain the kidney protective effects of empagliflozin in EMPA-REG OUTCOME.¹⁶⁻¹⁸ Of these, the effect of empagliflozin on diuresis (osmotic diuresis and natriuresis) could reflect a mechanism with particular importance for patients with HF and associated fluid overload.¹⁷ Increases in 24-hour urine volume of around 300 mL/d have been observed with empagliflozin after the first day of treatment but daily urine output returns to baseline within a few days.¹⁹ Furthermore, SGLT2 inhibition produces an acute but modest natriuresisrelated plasma volume contraction, which usually stabilizes within a few weeks, potentially helping to alleviate volume overload and reduce cardiac preload.¹⁶⁻¹⁸ Notably, a recent study showed that osmotic diuresis induced by SGLT2 inhibition has a distinctly different diuretic mechanism than that of other diuretic classes (eq, loop diuretics or thiazides).²⁰ By using a mathematical model derived from a clinical study in healthy individuals, the authors showed that SGLT2 inhibition resulted in greater electrolyte-free water clearance and, ultimately, in greater fluid clearance from the interstitial fluid space than from the circulation. The model predicted that an SGLT2 inhibitor reduces interstitial fluid volume by 3× as much as it reduces blood volume (480 versus 150 mL, respectively). In comparison, the predicted reduction in interstitial fluid volume with the loop diuretic bumetanide was only 66% of blood volume reduction (510 versus 780 mL, respectively).²⁰ It may thus be plausible that both plasma, as well as interstitial volume reduction, could contribute to a reduction in the risk of hospitalization for HF, which was observed in EMPA-REG OUTCOME.9 Furthermore, kidney organ decongestion and, in particular, reduction in kidney venous stasis may positively influence intraglomerular hemodynamics and may, at least in part, restore filtration physiology. In addition, the decrease in myocardial stretch, potentially mediated by contraction of plasma and cardiac interstitial volume, may have reduced cardiac arrhythmogenesis, which has also been proposed as a mechanism for the reduction in cardiovascular mortality seen in the trial.17

With regard to renal safety, no notable differences of the effect of empagliflozin were observed between HF subgroups in relation to kidney outcomes, nor with any AEs, serious AEs, and AEs leading to discontinuation of the therapy.⁹ Of note, the incidence of acute kidney-related AEs (both injury and failure) was higher in patients with versus without HF at baseline, regardless of treatment.

A prespecified eGFR slopes analysis has been reported in the overall population of EMPA-REG OUTCOME.⁸ The findings of the previous slopes analyses support the hypothesis that a hemodynamic effect of empagliflozin would lead to a decrease in intraglomerular pressure, an effect that, during long-term therapy, could result in preservation of kidney function.¹¹ As stated above, in the current slopes analysis the pattern of changes over 178 weeks with empagliflozin or placebo was similar for both HF subgroups, consistent with findings in the overall population, and suggesting that the effects of empagliflozin in kidney function over time are not affected by baseline HF status.

Several systemic and renal physiological effects, some already mentioned, may be contributing to the kidney-protective effects of empagliflozin observed in the current study. These include the alleviation of renal workload, in particular through via increased distal sodium delivery to the macula densa, activation of tubuloglomerular feedback and decreased hyperfiltration.¹⁷ In addition, empagliflozin may result in a potential improvement in renal oxygenation, resulting from a shift towards more favorable renal fuel energetics.^{21,22}

The current study has limitations. The subgroup of patients with HF in this post hoc analysis was modest (and comprises only ≈10% of the overall study population). In addition, the diagnosis of HF at baseline was based on investigator reports according to the narrow Standardized Medical Dictionary for Drug Regulatory Activities Query, in the absence of objective measures of biomarkers or cardiac function. Thus, it was also not possible to further assess the impact of reduced or preserved ejection fraction on kidney outcomes with empagliflozin.

Several randomized clinical trials are now underway to further investigate the potential role of SGLT2 inhibitors as a treatment specifically for HF, and these trials are enrolling HF patients with or without T2DM. These include EMPEROR-Reduced (Empagliflozin Outcome Trial in Patients With Chronic Heart Failure With Reduced Ejection Fraction) (URL: https://www. clinicaltrials.gov. Unique identifier: NCT03057977), EMPEROR-Preserved (URL: https://www.clinicaltrials. gov. Unique identifier: NCT03057951), Dapa-HF (URL: https://www.clinicaltrials.gov. Unique identifier: NCT03036124), DELIVER (Dapagliflozin Evaluation to Improve the Lives of Patients With Preserved Ejection Fraction Heart Failure) (URL: https://www.clinicaltrials. gov. Unique identifier: NCT03619213), and SOLOIST-WHF (Effect of Sotagliflozin on Cardiovascular Events in Patients With Type 2 Diabetes Post Worsening Heart Failure) (URL: https://www.clinicaltrials.gov. Unique identifier: NCT03521934).

In conclusion, the EMPA-REG OUTCOME trial previously reported that empagliflozin significantly reduced the risk of clinically relevant kidney outcomes⁸ and slowed progression of CKD.²³ Novel insights from this post hoc analysis add to this evidence by showing that these beneficial kidney effects of empagliflozin are also seen in the particular high-risk population of patients with concomitant HF. This suggests that reduction in intraglomerular pressure may have the potential to benefit long-term kidney outcomes in HF populations, and further clinical research addressing this hypothesis is currently underway.

ARTICLE INFORMATION

Received April 2, 2019; accepted April 26, 2019.

The Data Supplement is available at https://www.ahajournals.org/doi/suppl/10.1161/CIRCHEARTFAILURE.118.005875.

Correspondence

Javed Butler, MD, MPH, MBA, Department of Medicine, University of Mississippi Medical Center, 2500 N State St, Jackson, MS 39216. Email jbutler4@ umc.edu

Affiliations

Department of Medicine, University of Mississippi Medical Center, Jackson (J.B.). Institut Lorrain du Coeur et des Vaisseaux, Nancy, France (F.Z.). Division of Cardiology, St Michael's Hospital (D.F.) and Lunenfeld-Tanenbaum Research Institute, Mount Sinai Hospital (B.Z.), University of Toronto, Canada. Boehringer Ingelheim International GmbH, Ingelheim, Germany (A.K.-W., M.v.E., J.G., M.B.). Department of Diabetes, Central Clinical School, Monash University, Melbourne, Australia (A.K.-W.). Department of Medicine, Würzburg University Clinic, Germany (A.K.-W., C.W.). Boehringer Ingelheim Pharma GmbH & Co KG, Ingelheim, Germany (M.B.). Division of Nephrology & Hypertension, University of Utah, Salt Lake City (A.K.C.).

Acknowledgments

The EMPA-REG OUTCOME trial was funded by the Boehringer Ingelheim & Eli Lilly and Company Diabetes Alliance, which was involved in the initiation, design, and interpretation of the data and the decision to submit the research for publication. The authors thank the patients who participated in this trial, and Uwe Hehnke, Dipl Stat, Boehringer Ingelheim, who was involved in the initial data analysis. Medical writing assistance, supported financially by Boehringer Ingelheim, was provided by Charlie Bellinger of Envision Scientific Solutions, Horsham, United Kingdom, during the preparation of this article. The authors were fully responsible for all content and editorial decisions and were involved at all stages of article development and have approved the final version.

Sources of Funding

The EMPA-REG OUTCOME trial was sponsored by the Boehringer Ingelheim and Eli Lilly and Company Diabetes Alliance.

Disclosures

Dr Butler reports research support from the National Institutes of Health, Patient Centered Outcomes Research, and the European Union. He serves on the speakers bureau for Novartis, Janssen, and NovoNordisk. He serves as a consultant and serves on steering committee, clinical events committee, or data safety monitoring boards for Abbott, Adrenomed, Amgen, Array, AstraZeneca, Bayer, BerlinCures, Boehringer Ingelheim, Bristol Myers Squib, Cardiocell, CVRx, G3 Pharmaceutical, Innolife, Janssen, Lantheus, LinaNova, Luitpold, Medscape, Medtronic, Merck, Novartis, NovoNordisk, Relypsa, Roche, Sanofi, Stealth-Peptide, SC Pharma, V-Wave Limited, Vifor, and ZS Pharma. Dr Zannad is a consultant, speaker, or a member of an advisory board for Actelion, Amgen, AstraZeneca, Bayer, Boehringer, Boston Scientific, CEVA, CVRx, Vifor-Fresenius, GE Healthcare, J&J, KBP BioSciences, Livanova, Novartis, NovoNordisk, Pfizer, Quantum Genomics, Relypsa, Resmed, Roche, Takeda; ZS Pharma and the founder of CardioRenal. Dr Fitchett reports honoraria from Sanofi, Merck & Co, Amgen, AstraZeneca, Eli Lilly and Co, and Boehringer Ingelheim. Dr Zinman has received research grants awarded to his institution from Boehringer Ingelheim, AstraZeneca, and NovoNordisk, as well as honoraria from Janssen, Sanofi, Eli Lilly and Co, Boehringer Ingelheim, NovoNordisk, and Merck. Drs Koitka-Weber, von Eynatten, Zwiener, George, and Brueckmann are employed by Boehringer Ingelheim. Dr Cheung reports honoraria for advisory boards from Boehringer Ingelheim. Dr Wanner reports honoraria for advisory board and speaker activities from AstraZeneca, Boehringer Ingelheim, Eli Lilly, and Merck Sharp & Dohme.

REFERENCES

- Ambrosy AP, Fonarow GC, Butler J, Chioncel O, Greene SJ, Vaduganathan M, Nodari S, Lam CSP, Sato N, Shah AN, Gheorghiade M. The global health and economic burden of hospitalizations for heart failure: lessons learned from hospitalized heart failure registries. J Am Coll Cardiol. 2014;63:1123–1133. doi: 10.1016/j.jacc.2013.11.053
- 2. Ponikowski P, Voors AA, Anker SD, Bueno H, Cleland JG, Coats AJ, Falk V, González-Juanatey JR, Harjola VP, Jankowska EA, Jessup M, Linde C, Nihoyannopoulos P, Parissis JT, Pieske B, Riley JP, Rosano GM, Ruilope LM, Ruschitzka F, Rutten FH, van der Meer P; Authors/Task Force Members; Document Reviewers. 2016 ESC Guidelines for the diagnosis and treatment of acute and chronic heart failure: the task force for the diagnosis and treatment of acute and chronic heart failure of the European Society of Cardiology (ESC). Developed with the special contribution of the Heart Failure Association (HFA) of the ESC. *Eur J Heart Fail.* 2016;18:891–975. doi: 10.1002/ejhf.592
- 3. Seferović PM, Petrie MC, Filippatos GS, Anker SD, Rosano G, Bauersachs J, Paulus WJ, Komajda M, Cosentino F, de Boer RA, Farmakis D, Doehner W, Lambrinou E, Lopatin Y, Piepoli MF, Theodorakis MJ, Wiggers H, Lekakis J, Mebazaa A, Mamas MA, Tschöpe C, Hoes AW, Seferović JP, Logue J, McDonagh T, Riley JP, Milinković I, Polovina M, van Veldhuisen DJ, Lainscak M, Maggioni AP, Ruschitzka F, McMurray JJV. Type 2 diabetes mellitus and heart failure: a position statement from the Heart Failure Association of the European Society of Cardiology. *Eur J Heart Fail.* 2018;20:853–872. doi: 10.1002/ejhf.1170
- Fitchett D, Butler J, van de Borne P, Zinman B, Lachin JM, Wanner C, Woerle HJ, Hantel S, George JT, Johansen OE, Inzucchi SE; EMPA-REG OUTCOME® Trial Investigators. Effects of empagliflozin on risk for cardiovascular death and heart failure hospitalization across the spectrum of heart failure risk in the EMPA-REG OUTCOME® trial. *Eur Heart J.* 2018;39:363–370. doi: 10.1093/eurheartj/ehx511
- McAllister DA, Read SH, Kerssens J, Livingstone S, McGurnaghan S, Jhund P, Petrie J, Sattar N, Fischbacher C, Kristensen SL, McMurray J, Colhoun HM, Wild SH. Incidence of hospitalization for heart failure and case-fatality among 3.25 million people with and without diabetes mellitus. *Circulation*. 2018;138:2774–2786. doi: 10.1161/CIRCULATIONAHA.118.034986
- van Deursen VM, Urso R, Laroche C, Damman K, Dahlström U, Tavazzi L, Maggioni AP, Voors AA. Co-morbidities in patients with heart failure: an analysis of the European Heart Failure Pilot Survey. *Eur J Heart Fail.* 2014;16:103–111. doi: 10.1002/ejhf.30
- Zinman B, Wanner C, Lachin JM, Fitchett D, Bluhmki E, Hantel S, Mattheus M, Devins T, Johansen OE, Woerle HJ, Broedl UC, Inzucchi SE; EMPA-REG OUTCOME Investigators. Empagliflozin, cardiovascular outcomes, and mortality in type 2 diabetes. N Engl J Med. 2015;373:2117– 2128. doi: 10.1056/NEJMoa1504720
- Wanner C, Inzucchi SE, Lachin JM, Fitchett D, von Eynatten M, Mattheus M, Johansen OE, Woerle HJ, Broedl UC, Zinman B; EMPA-REG OUTCOME Investigators. Empagliflozin and progression of kidney disease in type 2 diabetes. *N Engl J Med.* 2016;375:323–334. doi: 10.1056/NEJMoa1515920
- Fitchett D, Zinman B, Wanner C, Lachin JM, Hantel S, Salsali A, Johansen OE, Woerle HJ, Broedl UC, Inzucchi SE; EMPA-REG OUT-COME® Trial Investigators. Heart failure outcomes with empagliflozin in patients with type 2 diabetes at high cardiovascular risk: results of the EMPA-REG OUTCOME® trial. *Eur Heart J.* 2016;37:1526–1534. doi: 10.1093/eurheartj/ehv728
- Zinman B, Inzucchi SE, Lachin JM, Wanner C, Ferrari R, Fitchett D, Bluhmki E, Hantel S, Kempthorne-Rawson J, Newman J, Johansen OE, Woerle HJ, Broedl UC. Rationale, design, and baseline characteristics of a randomized, placebo-controlled cardiovascular outcome trial of empa-

gliflozin (EMPA-REG OUTCOME™). *Cardiovasc Diabetol.* 2014;13:102. doi: 10.1186/1475-2840-13-102

- Wanner C, Heerspink HJL, Zinman B, Inzucchi SE, Koitka-Weber A, Mattheus M, Hantel S, Woerle HJ, Broedl UC, von Eynatten M, Groop PH; EMPA-REG OUTCOME Investigators. Empagliflozin and kidney function decline in patients with type 2 diabetes: a slope analysis from the EM-PA-REG OUTCOME Trial. J Am Soc Nephrol. 2018;29:2755–2769. doi: 10.1681/ASN.2018010103
- Cherney DZI, Zinman B, Inzucchi SE, Koitka-Weber A, Mattheus M, von Eynatten M, Wanner C. Effects of empagliflozin on the urinary albumin-to-creatinine ratio in patients with type 2 diabetes and established cardiovascular disease: an exploratory analysis from the EMPA-REG OUT-COME randomised, placebo-controlled trial. *Lancet Diabetes Endocrinol.* 2017;5:610–621. doi: 10.1016/S2213-8587(17)30182-1
- Cherney DZ, Perkins BA, Soleymanlou N, Maione M, Lai V, Lee A, Fagan NM, Woerle HJ, Johansen OE, Broedl UC, von Eynatten M. Renal hemodynamic effect of sodium-glucose cotransporter 2 inhibition in patients with type 1 diabetes mellitus. *Circulation*. 2014;129:587–597. doi: 10.1161/CIRCULATIONAHA.113.005081
- Skrtić M, Yang GK, Perkins BA, Soleymanlou N, Lytvyn Y, von Eynatten M, Woerle HJ, Johansen OE, Broedl UC, Hach T, Silverman M, Cherney DZ. Characterisation of glomerular haemodynamic responses to SGLT2 inhibition in patients with type 1 diabetes and renal hyperfiltration. *Diabetologia*. 2014;57:2599–2602. doi: 10.1007/s00125-014-3396-4
- Damman K, Gori M, Claggett B, Jhund PS, Senni M, Lefkowitz MP, Prescott MF, Shi VC, Rouleau JL, Swedberg K, Zile MR, Packer M, Desai AS, Solomon SD, McMurray JJV. Renal effects and associated outcomes during angiotensin-neprilysin inhibition in heart failure. *JACC Heart Fail*. 2018;6:489–498. doi: 10.1016/j.jchf.2018.02.004
- Butler J, Hamo CE, Filippatos G, Pocock SJ, Bernstein RA, Brueckmann M, Cheung AK, George JT, Green JB, Januzzi JL, Kaul S, Lam CSP, Lip GYH, Marx N, McCullough PA, Mehta CR, Ponikowski P, Rosenstock J, Sattar N, Salsali A, Scirica BM, Shah SJ, Tsutsui H, Verma S, Wanner C, Woerle HJ, Zannad F, Anker SD; EMPEROR Trials Program. The potential role and ra-

tionale for treatment of heart failure with sodium-glucose co-transporter 2 inhibitors. *Eur J Heart Fail*. 2017;19:1390–1400. doi: 10.1002/ejhf.933

- Heerspink HJ, Perkins BA, Fitchett DH, Husain M, Cherney DZ. Sodium glucose cotransporter 2 inhibitors in the treatment of diabetes mellitus: cardiovascular and kidney effects, potential mechanisms, and clinical applications. *Circulation*. 2016;134:752–772. doi: 10.1161/ CIRCULATIONAHA.116.021887
- Verma S, McMurray JJV, Cherney DZI. The metabolodiuretic promise of sodium-dependent glucose cotransporter 2 inhibition: the search for the sweet spot in heart failure. *JAMA Cardiol.* 2017;2:939–940. doi: 10.1001/jamacardio.2017.1891
- Heise T, Seewaldt-Becker E, Macha S, Hantel S, Pinnetti S, Seman L, Woerle HJ. Safety, tolerability, pharmacokinetics and pharmacodynamics following 4 weeks' treatment with empagliflozin once daily in patients with type 2 diabetes. *Diabetes Obes Metab.* 2013;15:613–621. doi: 10.1111/dom.12073
- Hallow KM, Helmlinger G, Greasley PJ, McMurray JJV, Boulton DW. Why do SGLT2 inhibitors reduce heart failure hospitalization? A differential volume regulation hypothesis. *Diabetes Obes Metab.* 2018;20:479–487. doi: 10.1111/dom.13126
- Mudaliar S, Alloju S, Henry RR. Can a shift in fuel energetics explain the beneficial cardiorenal outcomes in the EMPA-REG OUTCOME Study? A unifying hypothesis. *Diabetes Care*. 2016;39:1115–1122. doi: 10.2337/dc16-0542
- Muller ME, Pruijm M, Bonny O, Burnier M, Zanchi A. Effects of the SGLT-2 inhibitor empagliflozin on renal tissue oxygenation in non-diabetic subjects: a randomized, double-blind, placebo-controlled study protocol. *Adv Ther.* 2018;35:875–885. doi: 10.1007/s12325-018-0708-y
- Wanner C, Lachin JM, Inzucchi SE, Fitchett D, Mattheus M, George J, Woerle HJ, Broedl UC, von Eynatten M, Zinman B; EMPA-REG OUTCOME Investigators. Empagliflozin and clinical outcomes in patients with type 2 diabetes mellitus, established cardiovascular disease, and chronic kidney disease. *Circulation*. 2018;137:119–129. doi: 10.1161/CIRCULATIONAHA.117.028268